Author's response to reviews

Title: Smoking among pregnant women in Cantabria (Spain): trend and determinants of smoking cessation

Authors:

Silvia Palma (spalma@ujaen.es)
Rocio Perez-Iglesias (perez-iglesias@telefonica.net)
Rosa Pardo-Crespo (erodrigoc@ono.com)
Javier Llorca (llorcaj@unican.es)
Marcial Mariscal (mariscal@ujaen.es)
Miguel Delgado-Rodriguez (mdelgado@ujaen.es)

Version: 6 Date: 25 April 2007

Author's response to reviews:

Your suggestions have been followed: there are not ",." in the title. References are placed before punctuations signs. We now write "Authors' contributions", and the manuscript has been checked again.